

ANSWERING REVIEWERS

September 15, 2015



Dear Editor,

Please find enclosed the edited manuscript in word format (file name: 19519-Review.doc).

Title: Modelling the prevalence of hepatitis C virus amongst blood donors in Libya: An investigation of providing a preventive strategy

Author: Mohamed A Daw, Amira Shabash, Abdallah El-Bouzedi, Aghnya A Dau, Moktar Habas

Name of Journal: *World Journal Virology*

ESPS Manuscript NO: 19519

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated.

2 Revision has been made according to the suggestions of the reviewer:

We highlighted all the needed changes- blue colour.

3 Response to the Editor-in-Chief: We highlighted all the needed changes- blue colour.

4 References and typesetting were corrected.

Thank you again for publishing our manuscript in the *World Journal of Virology*.

Sincerely yours,

A handwritten signature in black ink, appearing to read 'MADAW', on a light-colored background.

Mohamed A Daw, MD, FTCDI